Author's response to reviews

Title: Clinical Assessment of Complementary Treatment with Qishen Yiqi Dripping Pills on Ischemic Heart Failure: A Randomized, Double-blind, Multicenter, Placebo-controlled Trial (CACT-IHF)

Authors:

Shuai Wang (wangshuai.hhht@163.com)
Yazhu Hou (mreleven@163.com)
Zhiqiang Zhao (quanmingzhao@126.com)
Xianliang Wang (xlwang1981@126.com)
Bin Li (xxlibin2006@163.com)
Shan Bin Soh (shanbin2@hotmail.com)
Jingyuan Mao (jymao@126.com)

Version: 2 Date: 5 January 2013

Author's response to reviews:

Shuai Wang1, Yazhu Hou2, Zhiqiang Zhao2, Xianliang Wang2, Bin Li2, Shan Bin Soh1, Jingyuan Mao2*

1Tianjin University of Traditional Chinese Medicine, 312 Anshan Western Road, Nankai District, Tianjin, China, 2Cardiovascular Department of The First Teaching Hospital of Tianjin University of Traditional Chinese Medicine, 314 Anshan Western Road, Nankai District, Tianjin, China.

Email: Shuai Wang- wangshuai.hhht@163.com; Yazhu Hou- mreleven@163.com;
Zhiqiang Zhao- quanmingzhao@126.com; Xianliang Wang- xlwang1981@126.com;
Bin Li- xxlibin2006@163.com; Shan Bin Soh- shanbin2@hotmail.com; Jingyuan Mao*- jymao@126.com